



# Review Review on Model Based Design of Advanced Control Algorithms for Cogging Torque Reduction in Power Drive Systems

Pierpaolo Dini \*,<sup>†</sup> and Sergio Saponara <sup>†</sup>

- Department of Information Engineering, University of Pisa, Via G. Caruso, 16, 56122 Pisa, Italy
- \* Correspondence: pierpaolo.dini@phd.unipi.it

+ These authors contributed equally to this work.

Abstract: This review of the state of the art aims to collect the description and main research results in the field of development and validation of control algorithms with the main purpose to solve the problem of cogging torque and main sources of electromagnetic torque ripple. In particular, we focus on electric drives for advanced and modern mechatronic applications such as industrial automation, robotics, and automotive applications, with special emphasis on work that exploits model-based design. A great added value of this paper is to explicitly show the operational steps required for the model-based design design of optimized control algorithms for electric drives where it is necessary to make up for electromagnetic torque oscillations due to the main sources of ripple, particularly cogging torque. The ultimate goal of this paper is to provide researchers approaching this particular problem with a comprehensive collection of the most effective solutions reported in the state of the art and also a summary for effectively applying the model-based design methodology.

**Keywords:** model-based design; simulation; mechatronics; dynamic systems; modeling; control theory; brushless motors; electric drives

# 1. Introduction

#### 1.1. Brief Introduction to MBD for Mechatronics

This state-of-the-art review focuses on a very specific topic such as the development of advanced algorithms for reducing cogging torque and major sources of electromagnetic torque disturbance in which MBD is exploited as a design paradigm.

As is well known, the use of MBD enables superior results to be achieved in the project phases, in which various aspects need to be evaluated upstream of the realization of a prototype, on which it will later be possible to concretize and refine the results.

The MBD approach has become particularly popular in the design of control algorithms for the automation of processes and dynamic systems of particular industrial interest [1,2].

Through the MBD approach, the risks associated with testing under critical operating conditions can be reduced, decreasing validation time and optimizing implementation steps [3].

So, MBD can be exploited to evaluate the choice of control techniques, numerical optimization criteria, operational approximations, modeling of phenomena at different levels of detail, computational complexity, cost-benefit ratio, preliminary evaluation of the required embedded platform, final integration on micro-controller/processor, etc. [4–6].

In addition, there are advanced tools implementing the MBD paradigm that make it easy to use for design and subsequent integration on electronic systems of interest, greatly speeding up implementation time.

Figure 1 schematically recalls the stages of the MBD paradigm.

Model-in-the-Loop (MIL) is the first step in the modeling-based development chain, in which both the control and monitoring algorithms and the physical plant or process are



Citation: Dini, P.; Saponara, S. Review on Model Based Design of Advanced Control Algorithms for Cogging Torque Reduction in Power Drive Systems. *Energies* **2022**, *15*, 8990. https://doi.org/10.3390/en15238990

Academic Editors: Quntao An, Bing Tian and Xinghe Fu

Received: 25 September 2022 Accepted: 22 November 2022 Published: 28 November 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



implemented through a set of mathematical equations and/or formalisms, in a simulation environment, such as MATLAB/Simulink.

Figure 1. Example of MBD steps for algorithms validation.

It is an essential step, as it allows us to make an exhaustive study of the system and of the interaction with the blocks that implement the algorithms, especially from the point of view of the operating space, verifying the functioning in various operating conditions and the consequences in their variations [7].

Software-in-the-Loop (SIL) is a process in which the actual Production Software Code, which in general can derive from developments prior to the adoption of MBD techniques, and therefore not necessarily self-generated code, is incorporated into a simulation environment that contains the mathematical models of the physical system.

This is done to allow for the inclusion of software features for which no templates exist or to allow for faster simulation runs, as part of the functional blocks are already written in low-level C / C ++ code. Therefore, SIL is defined as the inclusion of compiled production software code in a simulation model [8].

Processor-in-the-Loop (PIL) is the process in which the control and monitoring system is compiled and downloaded into an integrated target processor, ergo an embedded platform with micro-controller/processor with specifications consistent with that selected for the final mechatronic system and communicates directly with the model of the plant via standard communications such as Ethernet.

In this phase of the MBD validation process, hardware toolboxes are referred to as SW infrastructures that allow interaction between the simulation environment and the low-level environment, such as micro-controller or micro-processor. Among the most widely used in automotive and industrial automation is support for NXP's evaluation boards [9]. This allows designers to validate their control algorithm even in the absence of the physical process, by exploiting the simulated model, which interacts with the embedded code on the embedded system.

In this case, I/O devices are not used for communication, so unlike the RCP phase there is no verification of the interactions between the Embedded system and the physical process or part of it being tested [10].

Hardware-in-the-Loop (HIL) indicates the testing technique of electronic control units by connecting them to special benches that reproduce in a more or less complete way the electro-mechanical and electronic system of the process.

The purpose of the HIL tests is to use the benches to anticipate the checks on components, subsystems and systems already in the design and prototyping phase, without waiting for the availability of the final product for which they are intended.

In fact, the real components installed respond to the simulated signals as if they were operating in a real environment, since they are not able to distinguish signals coming from a physical environment from those produced by software models.

In this way, the HIL method makes it possible to reproduce the most diverse operating conditions with the benches and observing the behavior of the system and of the individual elements [11].

In this phase of validating control algorithms, one often has to deal with the problem of only having part of the physical process to be controlled. To increase the degree of validation with respect to the PIL phase, hardware emulators are often used. These devices, such as the widely used Speedgoat systems [12], are able to provide HW interfaces for system and process models, supplying currents and voltages instead of sensors and actuators of the final process. This allows designers to further test the validity of the algorithms by "getting closer" to the physical process with respect to PIL testing (as schematically reported in Figure 2).



**Figure 2.** MBD workflow in the specific case of design of control algorithms to reduce the cogging torque.

#### 1.2. Contributions

Many reviews of the state of the art regarding the reduction of Cogging Torque in Synchronous/Brushless motors by physical modification techniques can be found in the literature, but there are none regarding control techniques.

In this review, we focus on works that propose advanced control techniques for reducing cogging torque and major sources of torque disturbance. The development of the control techniques proposed in this work takes advantage of the MBD approach, which, as discussed above, is the most efficient and flexible. Compared with the reviews in the literature, our most relevant contributions are:

- Focus on the problem of cogging torque reduction by advanced control technique validated through MBD approach;
- Collect the most recent state-of-the-art work on this topic;

Critical analysis of the control techniques proposed by various authors, with the aim
of providing suggestions for implementation and improvement of the reported results.

#### 1.3. Paper Organization

The rest of the manuscript is organized into the following Sections: Section II explains the cogging torque problem in detail by reporting the main approaches and methods used for mathematical modeling in the context of control algorithm design; Section III gives details of mathematical modeling and integration in a simulation environment of the main components of electric drives in modern mechatronic systems; Section IV describes the most commonly used control techniques for solving the cogging torque problem by listing their advantages and disadvantages; and Section V gives final conclusions and a discussion of possible future developments.

#### 2. Cogging Torque's Details and Modeling

2.1. Reasons for Control Design in Cogging Torque Reduction

Given the enormous potential of this methodology, MBD is increasingly being used to increase the efficiency of individual components of an entire mechatronic process.

It can safely be said that the mechatronic processes of strongest industrial interest are physical electro-mechanical in nature, so among the most decisive components for maximum efficiency of the entire process are definitely electrical drives.

It can certainly be said that, by now, modern drives for advanced mechatronic applications mostly exploit Permanent Magnet Synchronous Motors (PMSM) and Brushless motors.

They have the most advantageous inertia/torque ratio, high efficiencies, physically robust characteristics, and are the most versatile for use at both very low and high speeds.

As is well known, among the inherent problems of using Synchronous/Brushless motors is torque ripple due to cogging torque, which causes undesirable oscillations, lowering performance both for low-speed applications where high accuracy in positioning control is required, and in speed control applications where unwelcome noise is generated.

There are some papers in the literature in which a very specific design is proposed with the aim of mitigating the effect of oscillations due to this physical phenomenon, but these have met with little success in practice [13–15].

This is mainly due to the fact that such solutions are based on physical modifications applicable only in some specific configurations, relating to Synchronous motors suitable mainly for applications in which moderate electromagnetic torque is required, since the underlying concept is to limit the amount of permanent magnets, arranging them so that the stator–rotor magnetic interaction is reduced, but effectively reducing the torque deliverable by the machine (see Figure 3).



**Figure 3.** Example of changing the arrangement of permanent magnets, valid only for surface magnets, called "rotor skewing".

The most promising results have been achieved by design of control algorithms, which are capable of compensating for cogging torque, which for all intents and purposes is an

additive disturbance signal with respect to electromagnetic torque, so control techniques are effective in compensating for it.

MBD being the best among advanced algorithm development and validation approaches, research in recent years sees the application of this paradigm for efficient resolution of this type of problem as well.

#### 2.2. Cogging Torque Modeling Approaches

As anticipated, cogging torque is an intrinsic phenomenon of permanent magnet synchronous motors and Brushless motors, both radial and axial flux, related to the magnetic interaction between the main parts of the electrical machine, such as stator and rotor.

The phenomenon is present in both synchronous motor configurations, with magnets superficial to the rotor and internal to the rotor iron, and Brushless, causing the same effect, i.e., electromagnetic torque disturbance.

The problem is intrinsic, as it depends on the unavoidable variation of the equivalent reluctance of the enclosing magnetic path of the magnetic flux concatenated with the stator iron through the air gap and produced by the permanent magnets themselves.

By manually rotating the rotor axis, it is possible to perceive this phenomenon as preferential angular positioning, at some angular positions "attractive" and at some angles "repulsive".

The phenomenon occurs even in the absence of power supply to the electrical machine, so it is interpreted as an additive disturbance with respect to the electromagnetic torque, due to the phase currents (and from the angular position).

In addition, due to the total symmetry of the ferromagnetic structure of electric machines, the alternation of the "repulsive" effect and the "attractive" effect due to the variation of the air gap, i.e., the magnetic reluctance, compensate each other along the entire round angle. This allows the intuition that cogging torque is a zero mean phenomenon.

In Figure 4, the concept of alternating "attractive" and "repulsive" phase is shown for a surface magnet synchronous motor configuration; the concept is the same for all configurations.



sense of rotation

Figure 4. Schematizing of the concept of alternating between "attractive" (a) and "repulsive" (b) phases.

It represents the behavior in the proximity of the preferential angular position, i.e., the position where naturally the concatenated magnetic flux tends to align the teeth of the stator slots with the magnets arranged on the rotor.

In the direction of rotation, the force generated by the concatenated magnetic flux (in the direction tangential to the rotor surface) alternates between positive and negative signs depending on whether the angular position is less or greater than the preferential position itself.

Obviously, the magnetic force is composed of a tangential component, which is the actual cause of cogging torque, and a normal component, which, however, is absorbed by the structure without any effect on the relative rotation between stator and rotor.

The application of advanced control techniques to reduce this inherent effect appears to be the best approach. Following the MBD design standard, a first step is certainly to make an estimate of the cogging torque by means of a mathematical model.

In the literature, the most widely used approaches that have been shown to be usable in the context of control algorithms, with the best results being closed form modeling and the online estimation.

Works in the literature that focus on mathematical modeling of cogging torque in closed form start from the equivalent energy representation of the machine, via an equivalent magnetic circuit. This highlights the angular position dependence of the stored magnetic power budget.

This relationship is further highlighted in the dependence of magnetic reluctance, due to the fact (described intuitively above) that the air gap varies its amplitude during relative rotation between stator and rotor [16-21].

Referring to Figure 5, it is possible to quantify the magnetic energy stored by the ferromagnetic structure as given in Equation (1).

$$W_m = \frac{1}{2} L_{eq} I^2 + \frac{1}{2} R_{eq}(\theta_m) \phi^2_{rs} + N I \phi_{rg}$$
(1)



Figure 5. Equivalent magnetic circuit for permanent magnet synchronous motors.

 $W_m$  denotes the magnetic energy stored by the electrical machine, both under energized and not energized conditions. With  $L_{eq}$  we denote the stator inductance of the equivalent circuit of the internal windings, *I* the phase current;  $R_{eq}$  the equivalent magnetic reluctance associated with the magnetic induction circuit at the mechanical air gap; and with *N* we denote the number of windings per stator phase. Finally,  $\phi_{rs}$  and  $\phi_{rg}$  denote the components of the magnetic flux concatenated through the air gap between the stator and rotor.

It follows that the cogging torque is defined as the derivative of magnetic energy with respect to angular position, in the absence of power, that is, at zero electric current in the machine windings. In Equation (2), the relationship between magnetic energy and cogging torque is given.

$$T_{cog} = -\frac{\partial W_m(I=0)}{\partial \theta_m} = -\frac{1}{2}\phi^2_{rs}\frac{\partial R_{eq}(\theta_m)}{\partial \theta_m}$$
(2)

In Equation (2), the term  $\frac{\partial W_m(I=0)}{\partial \theta_m}$  denotes the partial derivative of the magnetization energy with respect to the angular position of the rotor, calculated under the condition that the electrical machine is not powered by any energy source.

The exact expression of the cogging torque depends on the square of the magnetic flux intensity concatenated with the air gap  $\phi_{rs}$ , and the derivative of the equivalent magnetic reluctance, a function of angular position.

Note that  $\phi_{rs}$  depends on the physical characteristics of the permanent magnets, i.e., the magnetic material used, while  $R_{eq}$  depends substantially on geometrical factors, thus on the shape of the stator slots as well as on the arrangement of the permanent magnets themselves. Since this is all known information, this is an analytical representation in closed form.

This is the most widely used approach by researchers focusing on control algorithms because of the simplicity of integrating the cogging torque model, being precisely in closed form it can be integrated into the mechanical balance of rotor rotation.

Moreover, it is easy to validate through dedicated SW tools for finite element analysis that do not require the presence of the machine but only design information, which is often the case in advanced design activities typical of innovative companies exploiting R&D.

Another widely used approach, especially in the presence of the prototype electrical machine, is to use a test pod for direct measurement [22]. A typical test bench for cogging torque measurement is shown in Figure 6.



Figure 6. Typical cogging torque measurement test bench.

The typical measuring bench basically consists of two electric motors and a torque sensor (normally a torque transducer).

Through the first motor, which is called MASTER, we impose the motion of the second motor, called SLAVE, of which we are interested in recording the data inherent in the torque at the shaft by means of the torque transducer.

Since the Slave is "dragged" by the Master, its stator windings will not be current flowing, and this means that the torque measured by the torque transducer coincides with the Cogging Torque.

The important thing about this measurement is that it is made in the steady state of speed since the contribution of inertial actions due to acceleration transients must be avoided.

In fact, the measurement made in this way is the simplest procedure for obtaining a cogging torque profile that can be exploited in eventual modeling with a greater degree of accuracy describing the dynamic behavior of the machine and the design of a feedback law to compensate for such disturbance.

The result of the measurement is usually a set of data representing the measured cogging torque at different angular positions. This allows a closed-form model to be derived by approximation procedure, by harmonic analysis [23] or polynomial [24], as reported in Equation (3).

$$T_{cog} = \sum_{i=0}^{N_h} A_k cos \left( k \frac{\theta_m}{\theta_{cog}} + a_k \right) + B_k sin \left( k \frac{\theta_m}{\theta_{cog}} + b_k \right)$$
(3)

where  $A_k$ ,  $B_k$ ,  $a_k$  and  $b_k$  are harmonic development parameters, while  $\theta_{cog}$  is the angular period of the cogging pair, which depends on geometrical and construction factors, such as the number of stator slots and teeth, the number of permanent magnets and type of rotor iron arrangement (information known a priori).

Other works base the estimation of cogging torque on artificial intelligence models, neural networks and machine learning [25,26]. Few works, however, combine artificial intelligence models with control algorithms.

The main reason is certainly to reduce the overall computational complexity of the automatic cogging pair reduction system, and of course to reduce the storage space, which is usually the weak point of AI models compared to other approaches.

### 3. Synchronous Motor and Advanced Electric Power Drives Modeling

3.1. Fundamental Components of Electric Power Drives in Mechatronics

Another key step in the MBD approach is certainly accurate modeling of the electrical machine and drive components, including inverter, modulation system and implemented load.

A block diagram for the architecture of an electric drive for mechatronic applications is shown in Figure 7. The fundamental components of the system and their function are shown below.

- Static power converters: these are used to convert the electrical power regime to that which is useful for driving the electrical machine through the inverter. Configurations typically found in mechatronic systems of industrial interest are AC/DC converters to handle direct power from the mains or DC/DC converters to handle the DC bus level provided by a constant source, such as a Li-ion battery.
- Modulation system: conceptually separate from the system that integrates the control
  algorithm, but often implemented on the same embedded platform, it is the union of
  algorithm and HW interface for direct management of the inverter mosfets. The control
  algorithm typically provides a duty-cycle equivalent signal, which is converted into
  switching logic by the switching system, to manage the state of the inverter and power
  the electrical machine accordingly.
- **Inverter**: the electronic circuit that physically manages the (bidirectional) power transfer from the power source (DC side) to the electrical machine (AC side). The state of the inverter is decided by the modulation system. Conceptually, the inverter is a combinatorial logic network with 2<sup>3</sup> combinations (in the three-phase case).
- Control system: the embedded platform on which the logic related to the feedback control algorithm is integrated. In modern embedded systems it coexists with the modulation system and cooperates via SW drivers. In the sensorless configuration, such a system also integrates the angular position/velocity estimation algorithm by exploiting direct measurement of electric current (from inverter or motor phase).
- **Transduction system**: the set of devices for measuring the physical quantities of primary interest for managing the electric drive. Typically it consists of:
  - Temperature sensors, placed at "critical" points, such as the electrical machine and static converters, which are subject to high currents and therefore need to be monitored for losses due to the Joule effect.
  - Position and angular velocity sensors: placed on the rotor axis if we are talking about encoders/resolvers or placed inside the structure if we are talking about hall effect sensors.



Voltage/current sensors: typically placed on the inverter, since motor phases and inverter branches are necessarily traversed by the same current.

Figure 7. Typical electric drive architecture for mechatronic systems.

#### 3.2. Synchronous Motor Modeling including Cogging Torque

The mathematical model of the synchronous motor is typically derived by applying the unified theory of electrical machines, which states that any electrical machine is dynamically describable by an equivalent electrical and magnetic circuit [27].

Permanent-magnet synchronous motors of interest in mechatronic applications, where it is required to solve the cogging torque problem, are to date three-phase motors, so the following are the fundamental points of modeling for three-phase machines [28,29]. Note that the dynamic model is formally unchanged whether the machines have radial magnetic flux or axial magnetic flux.

In Equation (4), the dynamic balance of phase voltages of the electrical machine is given. The vector  $\vec{U}$  represents the triad of supply voltages, referred to the neutral of the equivalent circuit, the term  $R_s \vec{I}$  represents the losses in the stator circuit, and the term  $\frac{d\vec{Psi}}{dt}$  represents the voltage drop due to the phenomena of self and mute induction and the electromechanical energy conversion phenomenon, mainly due to the presence of the permanent magnets.

$$\vec{\mathcal{U}} = \begin{bmatrix} \mathcal{U}_a \\ \mathcal{U}_b \\ \mathcal{U}_c \end{bmatrix} = R_s \begin{bmatrix} I_a \\ I_b \\ I_c \end{bmatrix} + \frac{d}{dt} \begin{bmatrix} \Psi_a \\ \Psi_b \\ \Psi_c \end{bmatrix} = R_s \vec{I} + \frac{d\vec{\Psi}}{dt}$$
(4)

In Equation (5) are given the main components of the concatenated magnetic flux, denoted by the vector  $\vec{\Psi}$ . The vectors  $\vec{\Psi}_{AI}$  and  $\vec{\Psi}_{MI}$  denote the magnetic flux vectors generated by the phenomena of self and mutual induction, respectively, and the vector  $\vec{\Psi}_{PM}$  denotes the magnetic flux vector produced by the magnets concatenated at the air gap.

 $I_a$ ,  $I_b$  and  $I_c$  denote the components of the current vector (three-phase reference);  $U_a$ ,  $U_b$  and  $U_c$  denote the components of the phase voltage vector (three-phase axes); and

 $\Psi_a$ ,  $\Psi_b$  and  $\Psi_c$  are the components of the concatenated magnetic flux vector, taking into account both self and mutual induction at the mechanical air gap.  $R_s$  is the stator circuit electrical resistance.

$$\vec{\Psi} = \vec{\Psi}_{AI} + \vec{\Psi}_{MI} + \vec{\Psi}_{PM} = L_{eq}\vec{I} + k_{\Psi} \begin{bmatrix} \cos(p\theta) \\ \cos(p\theta - \frac{2\pi}{3}) \\ \cos\left(p\theta - \frac{4\pi}{3}\right) \end{bmatrix} = L_{eq}\vec{I} + k_{\Psi}\vec{C}(\theta)$$
(5)

Note that the magnetic flux vectors of self and mutual induction can be grouped in the term  $L_{eq}\vec{I}$ , where  $L_{eq}$  appropriately integrates the inductive coefficients of self and mutual induction.

In addition, the flux vector of permanent magnets  $\Psi_{PM}$  depends on the coefficient  $k_{\Psi}$ , indicating its intensity, which is a function of the ferromagnetic material used and the geometric arrangement, and depends formally on the angular position, as discussed above.

Combining Equation (4) and Equation (5), we derive the dynamical equilibrium of currents in the explicit form, in which the link between electrical ( $\vec{U}$  and  $\vec{I}$ ) and mechanical ( $\theta$  and  $\omega$ ) variables is evident. Equation (6) is the equation in vector form of the electrical dynamics of the Synchronous machine.

$$\vec{U} = R_s \vec{I} + L_{eq} \frac{d\vec{I}}{dt} + \frac{d\vec{\Psi}_{PM}}{dt} = R_s \vec{I} + L_{eq} \frac{d\vec{I}}{dt} + k_{\Psi} p \frac{d\theta}{dt} \frac{\partial \vec{C}(\theta)}{\partial \theta} = R_s \vec{I} + L_{eq} \frac{d\vec{I}}{dt} + \omega \vec{E}(\theta)$$
(6)

Equation (7) represents the balance of power terms, the meaning of which is explained below.

- The term  $P_{in}$  denotes the power absorbed by the machine and is formally defined as the scalar product between voltage vector  $\vec{U}$  and current vector  $\vec{I}$ .
- The term  $P_{joule}$  denotes the heating losses of the windings, which, as is known from physics, depend on the resistance of the circuit and the square of the modulus of the current, formally  $R_s < \vec{l}, \vec{l} >= R_s |\vec{l}|^2$ .
- The term  $P_{magnetic}$  represents the magnetic power stored in the equivalent circuit and due exclusively to self and mutual induction phenomena; formally  $L_{eq} < \frac{d\vec{l}}{dt}$ ,  $\vec{l} >$ .
- The term  $P_{mechanical}$  represents the mechanical power delivered by the machine, against electromechanical conversion (Lentz's law). For Synchronous motors, it depends on the angular position, via the counter-electro motion force  $\vec{E}(\theta)$  (more precisely its gradient), the electric current, and of course the angular speed.

$$P_{in} = \langle \vec{U}, \vec{I} \rangle = R_s \langle \vec{I}, \vec{I} \rangle + L_{eq} \langle \frac{d\vec{I}}{dt}, \vec{I} \rangle + \omega \langle \vec{E}(\theta), \vec{I} \rangle = P_{joule} + P_{magnetic} + P_{mechanic}$$
(7)

The most relevant term for dynamic modeling purposes is the contribution of the mechanical power output, which as we know is the product of electro-magnetic torque and rotor angular speed. Equation (8) shows the expression of the electro-magnetic torque, in the three-phase reference system.

$$T_{em} = <\vec{E}(\theta), \vec{I}> = -pk_{\Psi} \sum_{k=1}^{3} sin\left(p\theta - \frac{2(1-k)\pi}{3}\right) I_{k}$$
(8)

Equation (9) represents the dynamic rotational equilibrium for the rotor axis. Note that the second member of the equation represents the superposition of the contributions of electromagnetic torque  $T_{em}$ , Cogging torque  $T_{cog}$  and resistant torque associated with the mechanical load  $T_{load}$ . The parameters  $J_r$  and  $b_r$  represent the rotor moment of inertia and the coefficient of friction, respectively.

$$J_r \frac{d\omega}{dt} + b_r \omega = J_r \frac{d^2\theta}{dt^2} + b_r \frac{d\theta}{dt} = T_{em} + T_{cog} - T_{load}$$
(9)

An example of integration with block diagrams is shown in Figure 8, of the interaction between the set of differential equation representing the electrical balance of the phases of the electric motor and the differential equation representing the dynamic rotational balance of the rotor.



**Figure 8.** Implementation in Simulink environment of the dynamic interaction between electrical and mechanical subsystem.

Note how the output of one dynamic subsystem turns out to be the input for the other, defining a "recursive" dependence. Specifically, the electrical subsystem has the components of the voltage vector, position and angular velocity as inputs. The mechanical subsystem has as inputs, derived from the electrical subsystem, the currents (which in fact define the electromagnetic torque).

Figures 9 and 10 show the block diagrams, realized in the Simulink environment, of the dynamic subsystems describing the behavior of the electric motor. In Figure 9, note how the current dynamics of each phase is excited by the relative electro-motor control force component, which is a function of angular velocity and angular position.

Figure 10 shows that the mechanical equilibrium is activated from the electromagnetic torque, a function of phase currents and angular position. Note how the cogging torque model (red box in Figure 10) is inserted as an additive disturbance with respect to the electromagnetic torque.

To avoid the direct dependence of the electromagnetic torque expression on the angular position, the coordinate transformations are defined in Equation (10). The matrix  $B_{\alpha\beta}$  defines the Blondel transformation, while the matrix  $P_{dq}(\theta)$  formalizes the Park transformation.

$$B_{\alpha\beta} = \frac{2}{3} \begin{bmatrix} 1 & -\frac{1}{2} & -\frac{1}{2} \\ 0 & \frac{\sqrt{3}}{2} & -\frac{\sqrt{3}}{2} \\ \frac{1}{2} & \frac{1}{2} & \frac{1}{2} \end{bmatrix} P_{dq}(\theta) = \begin{bmatrix} \cos(\theta) & -\sin(\theta) & 0 \\ \sin(\theta) & \cos(\theta) & 0 \\ 0 & 0 & 1 \end{bmatrix}$$
(10)

Note that the Blondel transformation is basically the matrix form of the coordinate change due to the definition of space vector [30], which defines the decomposition of the three-phase vector  $\vec{X}_{abc} = [X_a, X_b, X_c]^T$  in the Gauss plane, as given in Equation (11).

$$X_{\alpha\beta} = \frac{2}{3} \left( X_a + X_b e^{j\frac{2\pi}{3}} + X_c e^{j\frac{4\pi}{3}} \right) = X_\alpha + jX_\beta$$
(11)

Combining the previous equation with the definition of homopolar component (typically used in electrical engineering)  $X_{om} = \frac{X_a + X_b + X_c}{3}$ , we obtain perfect equivalence with the Blondel transformation. Note that  $X_{om} = 0$  for symmetric and balanced systems (such as electric motors), defining a formally passive transformation.



**Figure 9.** Example realization in MATLAB/Simulink environment of the subsystem related to mechanical equilibrium.



**Figure 10.** Example realization in MATLAB/Simulink environment of the subsystem related to mechanical equilibrium.

Through the introduced coordinate transformations, we obtain the mathematical model that is exploited for the design of the control algorithms, given in Equation (12).

$$\begin{cases} U_d = R_s I_d + L_{eq} \frac{dI_d}{dt} - p\omega L_{eq} I_q \\ U_q = R_s I_q + L_{eq} \frac{dI_q}{dt} + p\omega (L_{eq} I_d + k_{\Psi}) \\ \frac{d\theta}{dt} = \omega \\ J_r \frac{d\omega}{dt} + b_r \omega = T_{em}^{dq} + T_{cog} - T_{load} \\ T_{em}^{dq} = \frac{3}{2} p k_{\Psi} I_q \end{cases}$$
(12)

#### 3.3. Modulation System in Advanced Power Drives

SVM (Space-Vector-Modulation) modulation is now the most widely used modulation technique, as it provides better performance than classical PWM (Pulse-Width-Modulation) based techniques.

As shown in Figure 11, reference is made to a classical configuration of an inverter with impressed voltage, connected to a three-phase load, which represents the equivalent circuit of the electrical machine in this specific case.

In the figure,  $V_ds$  has been used to denote the DC voltage, which is managed to modulate the inverter legs so as to energize the motor phases appropriately. The phase currents, which obviously coincide with those of the inverter branches, are referred to as  $I_{a,l}$   $I_b$  and  $I_c$ .



Figure 11. Topology of a two-level inverter with 3-phase load.

Note that the drive voltages of the inverter, denoted by  $V_{leg,a}$ ,  $V_{leg,b}$  and  $V_{leg,c}$ , are referred to the ground reference of the inverter circuit, unlike the phase voltages, denoted by  $V_a$ ,  $V_b$  and  $V_c$ , which are obviously referred to the star center of the stator windings.

Calling  $V_{no}$  the potential difference between the star-center reference n and the ground reference o, it is possible to find the expressions of the phase voltages of the electrical machine as a function of the inverter leg voltages, as expressed in Equations (13) and (14).

$$V_{leg,a} + V_{leg,b} + V_{leg,c} = V_a + V_b + V_c + 3V_{no} \longrightarrow V_{no} = \frac{V_{leg,a} + V_{leg,b} + V_{leg,c}}{3}$$
(13)

$$V_{a} = \frac{2}{3} V_{leg,a} - \frac{1}{3} (V_{leg,b} + V_{leg,c})$$

$$V_{b} = \frac{2}{3} V_{leg,b} - \frac{1}{3} (V_{leg,a} + V_{leg,c})$$

$$V_{c} = \frac{2}{3} V_{leg,c} - \frac{1}{3} (V_{leg,a} + V_{leg,b})$$
(14)

From these equations, the phase voltages corresponding to the possible state combinations of the inverter, resulting from modulation, are derived. Table 1 shows for each combination of inverter modulation signals the corresponding trio of phase voltages. Each triad actually corresponds to a sector of the SVM modulation.

Note that applying the definition of space vector to the eight triads corresponding to the possible eight states of the inverter, we obtain the vectors  $V_0$ , ...,  $V_7$ , which are positioned on the vertices of a hexagon in the Gauss complex plane.

These vectors, as shown in Figure 12, delimit the sectors with which the SVM modulation algorithm selects the most suitable state of the inverter, so that the desired voltage is averaged over the switching period  $T_{sw}$ .

From the control system, a vector of voltages (referenced to the star center) is generated typically in the Park dq reference system. After the inverse transformation to obtain the control triad in three-phase axes, the definition of space vector is applied via the Blondel transformation, the result of which is the vector  $V_{\alpha\beta}^*$ .

Space Vector	State Leg A	State Leg B	State Leg C	Va	V <sub>b</sub>	Vc
$V_0$	0	0	0	0	0	0
$V_1$	1	0	0	$\frac{2V_{dc}}{3}$	$-\frac{V_{dc}}{3}$	$-\frac{V_{dc}}{3}$
$V_2$	1	1	0	$\frac{V_{dc}}{3}$	$\frac{V_{dc}}{3}$	$-\frac{2V_{dc}}{3}$
$V_3$	0	1	0	$\frac{V_{dc}}{3}$	$\frac{2V_{dc}}{3}$	$-\frac{V_{dc}}{3}$
$V_4$	0	1	1	$-\frac{2V_{dc}}{3}$	$\frac{V_{dc}}{3}$	$\frac{V_{dc}}{3}$
$V_5$	0	0	1	$-\frac{V_{dc}}{3}$	$-\frac{V_{dc}}{3}$	$\frac{2V_{dc}}{3}$
$V_6$	1	0	1	$\frac{V_{dc}}{3}$	$-\frac{2V_{dc}}{3}$	$\frac{V_{dc}}{3}$
$V_7$	1	1	1	ŏ	0	ŏ

Table 1. Summary of the possible space vectors for three-phase inverter.

As shown in Figure 13, the control vector is instantaneously placed in one of the sectors of the modulation hexagon. The objective of SVM modulation is to reproduce the vector calculated by the control system, in mean value within the switching period, by partitioning the modulation vectors, as expressed in Equation (15).

$$\vec{V}_{\alpha\beta}^{*} = \frac{1}{T_{sw}} \left( T_1 \vec{V}_1 + T_2 \vec{V}_2 + T_0 \vec{V}_0 + T_7 \vec{V}_7 \right)$$
(15)

By  $T_1$ ,  $T_2$ ,  $T_0$ ,  $T_7$ , we have indicated the portions of the switching period in which the inverter is brought into a certain state by modulation. Note that the terms  $\frac{T_i}{T_{sw}}$  have the same meaning as the duty-cycle for the PWM technique.

Note, as schematized in Figure 14, that in order to integrate the SVM modulation, the triplet  $U_{abc}$  is transformed so that the vector  $V^*_{\alpha\beta}$  can be calculated. The modulus and angle are then extrapolated so that we can work out which sector is in the modulation hexagon. The selection of the sector and the calculation of the distribution times is done (for example) by a state machine.

Figure 15 shows the Simulink diagram of a possible implementation of an SVM algorithm for an ideal inverter (with instantaneous switching) in a simulation environment. In particular, it is shown how to realize the selection of each sector and the allocation of modulation states by means of a state machine realized with STATEFLOW.



**Figure 12.** Representation of the active and zero space vectors in the  $\alpha\beta$  plane, and its division in the modulation sectors (numbered by Roman numerals).







Figure 14. Simulink environment implementation of the SVM modulation technique.

### 3.4. Complete Dynamics Modeling including Cogging Torque

The last (but not least) fundamental component to be mathematically modeled is the electrical machine load. Mechatronic systems of industrial interest are mostly physical–electromechanical in nature. This allows the dynamics of the system implemented by permanent-magnet synchronous motors to be represented through the Lagrangian formalism [31].

$$L(q,\dot{q}) = K(q,\dot{q}) - U(q) = \sum_{i} K_{i}(q,\dot{q}) - \sum_{j} U_{j}(q)$$
  
=  $\sum_{i} \frac{1}{2} \Big( M_{i} |\vec{v}(q,\dot{q})|^{2} + \vec{\omega}_{i}^{T}(q,\dot{q}) J_{i}(q) \vec{\omega}_{i}(q,\dot{q}) \Big) - \sum_{j} M_{i} < \vec{g}, \vec{P}_{i} >$  (16)

Equation (16) reports the definition of the Lagrangian function *L* as the difference between the total kinetic energy of the system *K* and the potential energy *U*. The vector  $q = [q_1, ..., q_n]^T$  is composed of the Lagrangian variables, ergo the variables needed to describe the dynamics of the system.

The dynamics of the system can be derived by application of the variations method, from which the following equation is derived.

$$\frac{d}{dt}\frac{\partial L(q,\dot{q})}{\partial \dot{q}_{i}} - \frac{\partial L(q,\dot{q})}{\partial q} = Q_{i}^{T} = \sum_{i} < \vec{F}_{ext,i}, \frac{\partial \vec{P}_{i}(q)}{\partial q_{i}} > + \sum_{i} < \vec{T}_{ext,j}, \frac{\partial \vec{\alpha}_{i}(q)}{\partial q_{i}} >$$
(17)

The term  $Q_i$  represents the superposition of the non-conservative external effects;  $\vec{F}_{ext,i}$  and  $\vec{T}_{ext,j}$  are external, non-conservative forces and torques, respectively, applied to the system;  $\vec{P}_i$  is the point of application of the force, and  $\alpha_i$  is the angle subject to change under the action of the torque.

$$M(\vec{q})\ddot{\vec{q}} + C(\vec{q}, \dot{\vec{q}})\dot{\vec{q}} + G(\vec{q}) = \Gamma(\vec{q})\vec{F}_{act} - J^T(\vec{q})\vec{F}_{env}$$
(18)

Compactly, it is possible to rewrite the set of dynamic equations as in Equation (18), where  $M(\vec{q})$  is the inertia matrix,  $C(\vec{q})$  is the matrix of Coriolis terms,  $G(\vec{q})$  is the vector of gravitational and potential terms,  $\Gamma(\vec{q})$  is the matrix that maps the action of the actuators  $\vec{F}_{act}$  into the dynamics of the variables  $\vec{q}$ ,  $J(\vec{q})$  is the analytic Jacobian, and finally  $\vec{F}_{env}$  represents the effect of non-conservative external forces.



**Figure 15.** State machine representing the SVM modulation technique, integrated with STATEFLOW in Simulink environment.

Note that combining the model of the implemented system in Equation (18) with the mathematical model of the electrical machine in Equation (12) we obtain the final mathematical model, which allows us to design control algorithms for electro-mechanical systems that inherently takes into account the presence of the cogging torque.

### 4. Most Used Algorithms for Cogging Torque Reduction

The latest work in the literature, concerning the design of algorithms and control systems capable of compensating for cogging torque in Permanent Magnet Synchronous Motors dedicated to advanced mechatronic applications, has shown that the most effective techniques are precisely those in which the mathematical model of cogging torque can be handled as an internal detail in the overall model.

### 4.1. Model Predictive Control

In the MPC approach, the current control action is computed online rather than using a pre-computed, online, control law. A model predictive controller uses, at each sampling instant, the plant's current input and output measurements, the plant's current state, and the plant's model to:

- Calculate, over a finite horizon, a future control sequence that optimizes a given performance index and satisfies constraints on the control action;
- Use the first control in the sequence as the plant's input.

The MPC strategy is illustrated in Figure 16, where  $N_p$  is the prediction horizon, u(t + k | t) is the predicted control action at t + k given u(t). Similarly, y(t + k | t) is the predicted output at t + k given y(t).





We consider a discretized model of a dynamic system of the form of Equation (19).

$$\begin{aligned} \mathbf{x}(k+1) &= \mathbf{\Phi}\mathbf{x}(k) + \mathbf{\Gamma}\mathbf{u}(k) \\ \mathbf{y}(k) &= \mathbf{C}\mathbf{x}(k) \end{aligned} \tag{19}$$

where  $\Phi \in \mathbb{R}^{n \times n}$ ,  $\Gamma \in \mathbb{R}^{n \times m}$ , and  $C \in \mathbb{R}^{p \times n}$ . Applying the backward difference operator,  $\Delta x(k+1) = x(k+1) - x(k)$  is derived Equation (20).

$$\Delta \mathbf{x}(k+1) = \Phi \Delta \mathbf{x}(k) + \Gamma \Delta \mathbf{u}(k)$$
  

$$\Delta \mathbf{y}(k+1) = \mathbf{y}(k+1) - \mathbf{y}(k)$$
  

$$= \mathbf{C}\mathbf{x}(k+1) - \mathbf{C}\mathbf{x}(k)$$
  

$$= \mathbf{C}\Delta \mathbf{x}(k+1)$$
(20)

where  $\Delta u(k+1) = u(k+1) - u(k)$ . With further algebraic manipulation it is obtained the matrix form of augmented state form dynamics in Equation (21).

$$\begin{aligned} \mathbf{x}_{a}(k+1) \begin{bmatrix} \Delta \mathbf{x}(k+1) \\ \mathbf{y}(k+1) \end{bmatrix} &= \begin{bmatrix} \Phi & O \\ C\Phi & \mathbf{I}_{p} \end{bmatrix} \begin{bmatrix} \Delta \mathbf{x}(k) \\ \mathbf{y}(k) \end{bmatrix} + \begin{bmatrix} \Gamma \\ C\Gamma \end{bmatrix} \Delta \mathbf{u}(k) = \mathbf{\Phi}_{a}\mathbf{x}_{a}(k) + \mathbf{\Gamma}_{a}\Delta \mathbf{u}(k) \\ \mathbf{y}(k) &= \begin{bmatrix} O & \mathbf{I}_{p} \end{bmatrix} \begin{bmatrix} \Delta \mathbf{x}(k) \\ \mathbf{y}(k) \end{bmatrix} = \mathbf{C}_{a}\mathbf{x}_{a}(k) \end{aligned}$$
(21)

Suppose now that the state vector  $x_a$  at each sampling time, k, is available to us. Our control objective is to construct a control sequence as the follows.

$$\Delta \boldsymbol{u}(k), \Delta \boldsymbol{u}(k+1), \dots, \Delta \boldsymbol{u}(k+N_p-1)$$
<sup>(22)</sup>

where  $N_p$  is the prediction horizon, such that a given cost function and constraints are satisfied. The above control sequence will result in a predicted sequence of the augmented state vectors.

$$x_{a}(k+1 \mid k), x_{a}(k+2 \mid k), \dots, x_{a}(k+N_{p} \mid k)$$
(23)

which can then be used to compute predicted sequence of the plant's outputs.

$$y(k+1 \mid k), y(k+2 \mid k), \dots, y(k+N_p \mid k)$$
 (24)

Using the above information it is possible to compute the control sequence of Equation (22) and then apply u(k) to the plant to generate x(k + 1). Repeating the process again, using x(k + 1) as an initial condition to compute u(k + 1), and so on.

Most of the work in which MPC is used to reduce cogging torque, the approach of constructing the control sequence assuming known linearized model parameters is used Since the linearized model is derived by approximation from Equations (12) and (18), it contains the information about the extrapolated cogging model.

In the following, it is reported the approach to construct u(k) given x(k). Figure 17 represents schematically the MPC architecture. Using the plant model parameters and the measurement of  $x_a(k)$  is evaluated the augmented states over the prediction horizon successively applying the recursion formula, obtaining the set of equations in Equations (25).

$$\begin{aligned} \mathbf{x}_{a}(k+1 \mid k) &= \mathbf{\Phi}_{a} \mathbf{x}_{a}(k) + \mathbf{\Gamma}_{a} \Delta \mathbf{u}(k) \\ \mathbf{x}_{a}(k+2 \mid k) &= \mathbf{\Phi}_{a} \mathbf{x}_{a}(k+1 \mid k) + \mathbf{\Gamma}_{a} \Delta \mathbf{u}(k+1) \\ &= \mathbf{\Phi}_{a}^{2} \mathbf{x}_{a}(k) + \mathbf{\Phi}_{a} \mathbf{\Gamma}_{a} \Delta \mathbf{u}(k) + \mathbf{\Gamma}_{a} \Delta \mathbf{u}(k+1) \\ &\vdots \\ \mathbf{x}_{a}(k+N_{p} \mid k) &= \mathbf{\Phi}_{a}^{N_{p}} \mathbf{x}_{a}(k) + \mathbf{\Phi}_{a}^{N_{p}-1} \mathbf{\Gamma}_{a} \Delta \mathbf{u}(k) + \dots + \mathbf{\Gamma}_{a} \Delta \mathbf{u}(k+N_{p}-1) \end{aligned}$$
(25)



# Model Predictive Controller

Figure 17. State feedback model predictive controller.

In matrix form, it is possible to compact as in Equation (26).

$$\begin{bmatrix} \mathbf{x}_{a}(k+1 \mid k) \\ \mathbf{x}_{a}(k+2 \mid k) \\ \vdots \\ \mathbf{x}_{a}(k+N_{p} \mid k) \end{bmatrix} = \begin{bmatrix} \mathbf{\Phi}_{a} \\ \mathbf{\Phi}_{a}^{2} \\ \vdots \\ \mathbf{\Phi}_{a}^{N_{p}} \end{bmatrix} \mathbf{x}_{a}(k) + \begin{bmatrix} \Gamma_{a} \\ \mathbf{\Phi}_{a}\Gamma_{a} & \Gamma_{a} \\ \vdots \\ \mathbf{\Phi}_{a}^{N_{p}-1}\Gamma_{a} & \cdots & \Gamma_{a} \end{bmatrix} \begin{bmatrix} \Delta u(k) \\ \Delta u(k+1) \\ \vdots \\ \Delta u(k+N_{p}-1) \end{bmatrix}$$
(26)  
$$\mathbf{Y} = \begin{bmatrix} \mathbf{y}(k+1 \mid k) \\ \mathbf{y}(k+2 \mid k) \\ \vdots \\ \mathbf{y}(k+N_{p} \mid k) \end{bmatrix} = \begin{bmatrix} C_{a}\mathbf{x}_{a}(k+1 \mid k) \\ C_{a}\mathbf{x}_{a}(k+2 \mid k) \\ \vdots \\ C_{a}\mathbf{x}_{a}(k+N_{p} \mid k) \end{bmatrix}$$
$$= \begin{bmatrix} C_{a}\Phi_{a} \\ C_{a}\Phi_{a}^{2} \\ \vdots \\ \mathbf{y}_{a}(k) \end{bmatrix} \mathbf{x}_{a}(k)$$
(27)

$$\begin{bmatrix} \vdots \\ C_a \Phi_a^{N_p} \end{bmatrix} + \begin{bmatrix} C_a \Gamma_a \\ C_a \Phi_a \Gamma_a & C_a \Gamma_a \\ \vdots & \ddots \\ C_a \Phi_a^{N_p-1} \Gamma_a & \cdots & C_a \Gamma_a \end{bmatrix} \begin{bmatrix} \Delta u(k) \\ \Delta u(k+1) \\ \vdots \\ \Delta u(k+N_p-1) \end{bmatrix} = \mathbf{W} \mathbf{x}_a(k) + \mathbf{Z} \Delta u(k)$$

Equation (27) describes the dependence of the sequence of future output vectors, in the prediction horizon, as a function of the increased state vectors, in that prediction horizon, and as a function of the change in the control vector. This equation is decisive in solving the optimization problem, which is formalized in the following.

$$J(\Delta \boldsymbol{U}) = \frac{1}{2} (\boldsymbol{r}_p - \boldsymbol{Y})^\top \boldsymbol{Q} (\boldsymbol{r}_p - \boldsymbol{Y}) + \frac{1}{2} \Delta \boldsymbol{U}^\top \boldsymbol{R} \Delta \boldsymbol{U}$$
$$\boldsymbol{Y}^{\min} \leq \boldsymbol{Y} \leq \boldsymbol{Y}^{\max}$$
$$\boldsymbol{U}^{\min} \leq \boldsymbol{U} \leq \boldsymbol{U}^{\max}$$
(28)

Equation (28) represents the cost functional that is minimized to derive the predictive sequence of optimal control, according to the criterion defined precisely by  $J(\Delta u)$ .

$$\Delta \boldsymbol{u}(k) = \overbrace{\left[ I_m \ \boldsymbol{O} \ \cdots \ \boldsymbol{O} \right]}^{N_p \text{ block matrices}} \left( \boldsymbol{R} + \boldsymbol{Z}^\top \boldsymbol{Q} \boldsymbol{Z} \right)^{-1} \boldsymbol{Z}^\top \boldsymbol{Q} \left( \boldsymbol{r}_p - \boldsymbol{W} \boldsymbol{x}_a \right)$$
(29)  
$$= \boldsymbol{K_r} \boldsymbol{r}_p - \boldsymbol{K_x} \Delta \boldsymbol{x}(k) - \boldsymbol{K_y} \boldsymbol{y}(k)$$

For the control vector to be optimal and meet all operational constraints on the control and output variables, minimization of the functional is associated with an optimizer based, for example, on gradient descent algorithm or Newtonian methods.

As can be understood from the procedure for constructing the control vector, MPC actually acts as a LQR (Linear-Quadratic-Regulator) controller in which it is possible to explicitly insert operational constraints on the controlled and control variables.

This represents an enormous advantage over the classical optimal control approach, as the control system solution can take into account the realistic constraints of the physical system to be controlled, and furthermore, there is the certainty that the control vector provides (sub-)optimal action in each of its components, being the result of evaluation over successive time windows.

On the other hand, MPC requires a very high computational effort, which is not always suitable for micro-controller-based embedded systems.

This is because at each step of the algorithm, an internal simulation is run, which exploits the mathematical model of the physical process to predict behavior and to be able to calculate a control vector, which will be optimal over that simulated time window.

Thus, the system on which the algorithm is integrated must be able to simulate  $N_p$  steps of the process model, within a single numerical integration step.

Another small disadvantage is that MPC is still an optimal control technique based on the mathematical model, so it is very sensitive to model uncertainties.

It is, however, very much used in the context of cogging torque reduction, as the ability to correctly model the electrical machine and the cogging phenomenon is quite optimized, leaving little room for model uncertainties that the feedback control may not be able to handle [32–35].

#### 4.2. Feedback Linearization Control

Another widely used control technique is Feedback Linearization Control (FLC), which takes advantage of differential geometry theory to directly handle model dynamics in a nonlinear form.

In order to apply the control vector design technique by Feedback Linearization, it is necessary that the mathematical model of the mechatronic system of interest be in the state form called "similar in control", as given in Equation (30), where the vector field f(x) is called drift vector, while the  $g_k(x)$  is the control vectors field.

$$\frac{dx}{dt} = F(x, u) = f(x) + \sum_{k} g_{k}(x)u_{k} ; x_{0} = x(t_{0})$$

$$y_{l} = H_{l}(x, u) = h_{l}(x)$$
(30)

Being able to handle the inherent non-linearities of the mathematical model in the generic state form of Equation (30), the resulting control law by definition contains information about the cogging pair, via the model chosen to represent it.

$$\frac{dh_i}{dt} = \frac{\partial h_i}{\partial x} \frac{\partial x}{\partial t} = \frac{\partial h_i}{\partial x} \left( \vec{f} + \sum_{k=1}^{n_u} \vec{g}_k u_k \right) = \frac{\partial h_i}{\partial x} \vec{f} = L_f h_i$$

$$\dots$$

$$\frac{d^{\mu_i} h_i}{dt^{\mu_i}} = \frac{d}{dt} \left( L_f^{\mu_i - 1} \right) = \frac{\partial}{\partial x} \left( L_f^{\mu_i - 1} \right) \left( \vec{f} + \sum_{k=1}^{n_u} \vec{g}_k u_k \right) = L_f^{\mu_i} h_i + \sum_{k=1}^{n_u} L_{g_k} L_f^{\mu_i - 1} h_i u_k$$
(31)

The operative procedure for applying feedback linearization involves deriving the control outputs as many times as necessary to obtain expressions in which the contributions

of the control variables appear. In Equation (31), the expression of the derivative  $\mu_i$ th for the *i*th control output is given in general form. We denote the relative degree for the output *i*th by  $\mu_i$ , with the condition that  $\sum \mu_i \leq n_x$ .

$$\begin{bmatrix} L_{f}^{\mu_{1}}h_{1} \\ L_{f}^{\mu_{2}}h_{2} \\ \dots \\ L_{f}^{\mu_{l}}h_{l} \end{bmatrix} + \begin{bmatrix} L_{g_{1}}L_{f}^{\mu_{1}-1}h_{1} & \dots & L_{g_{m}}L_{f}^{\mu_{1}-1}h_{1} \\ L_{g_{1}}L_{f}^{\mu_{2}-1}h_{2} & \dots & L_{g_{m}}L_{f}^{\mu_{2}-1}h_{2} \\ \dots & \dots & \dots \\ L_{g_{1}}L_{f}^{\mu_{l}-1}h_{l} & \dots & L_{g_{m}}L_{f}^{\mu_{1}-l}h_{l} \end{bmatrix} \vec{u} = \vec{v} = \vec{\Gamma}(\vec{x}) + \Sigma(\vec{x})\vec{u}$$
(32)  
$$\vec{u} = \Sigma^{+}(\vec{x})\left(\vec{v} - \vec{\Gamma}(\vec{x})\right) + (I - \Sigma^{+}(\vec{x})\Sigma(\vec{x}))\lambda$$

Proceeding with the derivation of each control output, in general we obtain the matrix expression given in Equation (32), where with  $L_{g_i}L_f h_j(\vec{x}) = \frac{\partial}{\partial \vec{x}} \left( \frac{\partial h_j}{\partial \vec{x}} \vec{f} \right) \vec{g}_i$  is the Lie derivative of the function  $h_j$  in the directions of the vector fields  $\vec{f}$  and  $\vec{g}_i$ , respectively.

$$\dot{\vec{z}} = \begin{bmatrix} A_1 & 0 & 0 & \dots & 0 \\ 0 & A_2 & 0 & \dots & 0 \\ \dots & \dots & \dots & \dots & \dots \\ 0 & 0 & 0 & \dots & A_l \end{bmatrix} \vec{z} + \begin{bmatrix} B_1 & 0 & \dots & 0 \\ 0 & B_2 & \dots & 0 \\ \dots & \dots & \dots & \dots \\ 0 & 0 & \dots & B_l \end{bmatrix} \vec{v}$$

$$\vec{z} = \begin{bmatrix} \vec{z}_1 \\ \vec{z}_2 \\ \dots \\ \vec{z}_l \end{bmatrix} ; \ \vec{z}_i = \begin{bmatrix} h_i \\ L_f h_i \\ \dots \\ L_f^{\mu_i - 1} \end{bmatrix} ; \ A_i = \begin{bmatrix} 0 & 1 & 0 & 0 & \dots & 0 \\ 0 & 0 & 1 & 0 & \dots & 0 \\ \dots & \dots & \dots & \dots & \dots \\ 0 & 0 & 0 & 0 & \dots & 1 \\ 0 & 0 & 0 & 0 & \dots & 0 \end{bmatrix} ; \ B_i = \begin{bmatrix} 0 \\ 0 \\ \dots \\ 0 \\ 1 \end{bmatrix}$$

$$(33)$$

The feedback linearization procedure results in a change of coordinates in which the dynamics of the new state vector is linear, as shown in Equations (33). Note that the state representation in addition to being linear represents a chain of integrators, consequently the pair ( $A_i$ ,  $B_i$ ) turns out to be completely reachable/controllable  $\forall i$ .

$$\vec{\tilde{z}} = \begin{bmatrix} \vec{z} \\ \vec{\sigma}(\vec{x}) \end{bmatrix}$$
 with  $\vec{\sigma}(\vec{x})$  s.t  $\frac{\partial \vec{z}}{\partial \vec{x}} \neq 0$  (34)

In general, a change of basis with reduction of the cardinality of the state space is obtained. Formally, this requires that certain mathematical conditions are fulfilled, which define the criterion of zero dynamics [36]. The first condition is given in Equation (34), where we impose that the Jacobian of the base change is formally lawful (invertible). This means that the completion  $\vec{\sigma}(\vec{x}) \in R^{n_x - \sum \mu_i}$  will have to be constructed appropriately to satisfy the condition.

$$\dot{\vec{\sigma}}(\vec{x}) = \vec{\gamma}(\vec{z},\vec{\sigma}) + \vec{\xi}(\vec{z},\vec{\sigma})v \tag{35}$$

The completion vector  $\vec{\sigma}(\vec{x})$  will in general have a non-linear state representation, dependent on both the new and old state variables and also on the new control vector, as given in Equation (35).

$$L_{g_k}\sigma_i = 0 \ \forall i,k \ \to \ \vec{\sigma}(\vec{x}) = \vec{\gamma}(\vec{z},\vec{\sigma}) \ \to \ \vec{\sigma}(\vec{x}) = \vec{\gamma}(0,\vec{\sigma}) \ A.S. \tag{36}$$

The second condition, given in Equation (36), serves to eliminate the contribution of the new control vector v from the dynamics of the vector  $\vec{\sigma}$ . In this way, it is possible to implement the zero-dynamics criterion, which guarantees us that even in the case of a reduction of the cardinality of the state space, the vector v constructed by means of a feedback linearization technique is able to asymptotically stabilize even the state variables that serve only to make the base change formally lawful. In the above equation, "A.S" means asymptotically stable.

The FLC-based control synthesis approach is schematized in Figure 18. Note that two control "loops" can be identified, one external and one internal. The inner one consists of the nonlinear terms based on the Lie derivative calculus, shown earlier, which is intended to map the control action of the outer loop into the representation space of the nonlinear dynamical system. The outer loop exploits the change of coordinates to the linear representation domain. In fact, the outer loop is dedicated to reducing the error between the reference and the output of the system, and is realized by linear control techniques.



Figure 18. Schematic representation of the internal structure of FLC controller.

As can be seen from the operating procedure for constructing the control vector using the FLC technique, the greatest advantage of this method is that it can handle highly non-linear dynamic models, being able to include very realistic details in the model itself, such as the cogging torque in this particular case.

Some inconvenience arises when the relative degree of the system is much less than the cardinality of the state space associated with the starting model. In this case, one must choose a base change completion that is invertible, so as to make it formally acceptable.

However, it can be said that in practical cases it is always possible to construct performance outputs such that they return a relative degree equal to (or slightly less than) the dimension of the state vector of the initial non-linear model.

For electro-mechanical systems of industrial interest, such as in robotics and industrial automation, dynamic system models are also found to have suitable formal properties for the application of this control methodology, which is why it is one of the most widely used control techniques by designers of algorithms and control systems [37–42].

### 4.3. Resonant Control

Control techniques based on the concept of resonance are often used by designers to resolve and attenuate the presence of periodic disturbances, whose pulsation/frequency can be measured or estimated with reasonable accuracy. In fact, resonant type controllers are well suited to the rejection of additive disturbances in sinusoidal form, as is the case with cogging torque.

This is the reason why some works in the literature are based on the use of resonant techniques, such as PR (Proportional Resonant) and PIR (Proportional Integral Resonant). Since cogging torque is an additive disturbance compared to electro-magnetic disturbance with periodic comparing often analyzed by harmonic models, it proves to be a rather efficient technique under certain types of conditions.

The transfer function shown in Equation (37) represents the Laplace transform domain implementation (for SISO = Single-Input-Single-Output systems) of a PR (Proportional-Resonant) type controller.

$$G_C(s) = K_p + \frac{2K_i s}{s^2 + \omega_0^2}$$
(37)

 $K_p$  denotes the proportional gain,  $K_i$  denotes the gain obtained when the additive disturbance on the output has exactly the pole pulsation of the transfer function  $\omega_0$ , known as the resonance pulsation or maximum feedback gain.

In Equation (38) is shown the version that is most often taken into account for integration convenience. Since the previous equation contains complex conjugate poles, in order to facilitate discretization and thus implementation in low-level code, a damping factor is often inserted, which also improves the controller's response, making it less abrupt and consequently requiring less control effort.

$$G_{\rm C}(s) = G_{Cp}(s) + G_{Cr}(s) = K_p + \frac{2K_i\omega_c s}{s^2 + 2\omega_c s + \omega_0^2}$$
(38)

The pulse  $\omega_c$  is called the cut-off pulse and differs from  $\omega_0$  obviously, in practice  $\omega_c < \omega_0$ . Having inserted a damping factor, the disturbance pulse and the resonant filter within the control now differ, so the feedback gain will not be maximum at the same time as the disturbance pulse. Theoretically, disturbance rejection is less efficient than the ideal PR case, but in practice we find that performance is virtually equivalent, also benefiting from smoother control action.

Equation (39) represents a possible implementation of the resonant PR controller by discretization. In particular, by simply using a bi-linear transformation, one obtains a discrete system that is not strictly proper (which in the discrete domain continues to make physical sense).

$$G_{Cr}(z) = \frac{Y(z)}{E(z)} = \frac{a_1(1-z^{-2})}{b_0+b_1z^{-1}+b_2z^{-2}} \quad \text{with} \quad \begin{aligned} a_1 &= 4K_i T_s \omega_c \\ b_0 &= T_s^2 \omega_0^2 + 4T_s \omega_c + 4 \\ b_1 &= 2T_s^2 \omega_0^2 - 8 \\ b_2 &= T_s^2 \omega_0^2 - 4T_s \omega_c + 4 \end{aligned}$$
(39)

Applying the inverse Z-transform yields the finite difference equation (or recursive equation) shown in Equation (40). The implementation on a microcontroller therefore requires memory to be kept of two steps prior to the current one.

$$y(k) = \frac{1}{b_0} [a_1 \cdot e(k) - a_1 \cdot e(k-2) - b_1 \cdot y(k-1) - b_2 \cdot y(k-2)]$$
(40)

Figure 19 shows how to realize a block diagram equivalent to the PR controller in the form of a Z-transformed transfer function, Simulink environment, using only 'native' blocks.

The tuning of resonant controller parameters is quite easy to achieve in practice, which is why it is among the most frequently used techniques by control algorithm designers for this type of problem. The most commonly used technique is that of forced oscillations, often accompanied by the classic Ziegler–Nichols table [43–48]. Due to the ease of development and integration on an embedded platform, it is also one of the most widely used control techniques in power and control electronics in general [49–51], not only in electrical drives.

#### 4.4. Others Control Techniques

Few other works in the literature exploit adaptive control techniques for cogging torque reduction in servo drives for mechatronic systems [52–55].

Adaptive control in general can be divided into direct approach and indirect approaches:

Direct Adaptive Control: in this case, the mathematical model of the physical system is taken as known and the control action is adapted by going to change its online parameters;

• **Indirect Adaptive Control**: in this case, a simplified model, such as a linearized version, is used as a basis, going to modify its parameters based on the error between the actual measurements and the model's prediction, also modifying a posterior the controller parameters that must be adapted accordingly.



Figure 19. Proportional resonant controller implementation example.

Figure 20 shows the principle diagrams of both approaches of adaptive control. In both cases, it is possible to derive a control law that incorporates cogging torque information, reason why it was deemed appropriate to include this approach among the techniques that can be practically used.



**Direct Adaptive Control Architecture** 

Figure 20. A schematic representation of direct vs. indirect adaptive control.

In the case of direct adaptive control, one of the cogging pair models previously discussed can be exploited. To exploit online adaptive paradigms usually use error-corrector

learning algorithms such as gradient descent, which work best on linear models. So in the case of indirect adaptive control, if the estimation of cogging torque-related parameters is based on the linearized model, the compensation may not be as efficient as in the direct case.

Very few works in the literature exploit the Sliding Mode control technique instead, which is classified as variable structure control [56,57]. It is based on the mathematical concept of region of attraction so as to derive a control law that typically depends on the sign of the hyperplane describing that region of attraction.

The main problem with this control technique is that it works well when the system model has slight non-linearities and a relative degree (the number of times the output must be derived to obtain an expression containing the control variables) of unity. It is used side-by-side with other types of controllers because, in any case, it is based on a mathematical model of the motor in which it is actually possible to incorporate the cogging torque information to be compensated by the resulting control action.

#### 4.5. Final Considerations

4.5.1. Model Predictive Control

Main advantages:

- Easy way of dealing with constraints on controls and states;
- High performance controllers and accurate;
- No need to generate solutions for the whole time-horizon;
- Flexibility: managing any model, any optimal objective function;
- Includes cogging torque information directly in the control law.

Main disadvantages:

- High computational complexity (poorly suited to typical embedded systems);
- Problems of robustness with respect to uncertainties in the reference mathematical model;
- Strict dependence on the dynamic model.

# 4.5.2. Feedback Linearization Control

Main advantages:

- Management of highly non-linear models;
- Joint use of other highly effective techniques on stationary linear systems;
- Fully algebraic control law (no on-line recursion required);
- Includes information on cogging torque through the most accurate model possible (optimizing compensation);
- Relative simplicity of implementation;
- Robust respect to the model parameters uncertainty.

Main disadvantages:

- Strict dependence on the mathematical modeling;
- Does not inherently manage operational constraints on dynamic variables.

### 4.5.3. Resonant Controller

Main advantages:

- Good ability to handle sinusoidal disturbances (as the cogging torque);
- Relative easy to tune;
- Simple to implement in embedded system.

Main disadvantages:

- Strongly dependent on the operating condition under which the control coefficients are chosen;
- Does not exploit completely the cogging torque modeling.

4.5.4. Adaptive and Sliding Mode Controllers

The main advantage is related to a relatively small computational complexity (more about the sliding mode technique).

The main disadvantages are:

- Inherent problems of convergence and numerical stability;
- Partial or ineffective use of the cogging torque model.

# 5. Conclusions

This paper proposed a discussion of how to apply the MBD approach in the context of the design of control systems and algorithms for cogging torque reduction, a rather hot topic for advanced mechatronic applications where high efficiency and performance is required, and especially a topic of current research on the control of permanent magnet electric machines.

During this review of the state of the art, the problem of torque cogging was analyzed in detail justifying the importance of addressing this issue through control algorithms in order to realize flexible and scalable solutions for various mechatronic applications and systems.

Popular mathematical modeling methodologies were presented with a view to being able to exploit cogging torque information directly in the design of control algorithms.

It was shown how to carry out mathematical modeling and in a simulated environment, of the components of the mechatronic system: the electric motor, presenting an overview of the dynamic model that integrates the cogging torque into the mechanical equilibrium; the modulation system, now based almost exclusively on SVM technique, presenting the theoretical principles and a possible integration in a simulated environment, with a view to successfully apply MBD validation of the control algorithms, in the most realistic simulation possible; the mechanical load part and the connection with the dynamics of the machine to derive the most complete mathematical model possible.

The most popular control techniques for this type of problem were then analyzed, in particular to those presented in the most recent literature articles and which showed the most promising results. The purpose is to give an overview of the algorithms by discussing their advantages and disadvantages, which were summarized in a table.

The analysis led to the statement that MPC and FLC are certainly the most efficient control algorithms. On the one hand, MPC incorporates the ability to handle the operational constraints of the problem, ensuring that the control action is physically achievable by the system components, and on the other hand, FLC's ability to handle inherently non-linear systems, the complete information about the cogging pair and to have an inherent robustness to parametric uncertainties.

It is expected that future work will exploit the strengths of these two control techniques, taking advantage of FLC's architecture characterized by a linear outer loop and nonlinear inner loop. Since MPC works excellently with discretized linear systems, an innovative idea could be to transform the mathematical model representation that includes the cogging pair, with nonlinear dynamics, into an equivalently linear model, and to apply MPC in the outer loop, further ensuring that the final control action respects the physical constraints of the drive.

**Author Contributions:** Conceptualization, S.S. and P.D.; methodology, S.S. and P.D.; software, P.D.; validation, S.S. and P.D.; writing-original draft preparation, S.S. and P.D.; writing—review and editing, S.S. and P.D. All authors have read and agreed to be published version of the manuscript.

Funding: This work was funded by MIUR through project Dipartimenti di Eccellenza Crosslab.

**Data Availability Statement:** The authors confirm that the data supporting the findings of this study are available within the article.

Conflicts of Interest: The authors declare no conflict of interest.

# References

- 1. Akundi, A.; Lopez, V. A Review on Application of Model Based Systems Engineering to Manufacturing and Production Engineering Systems. *Procedia Comput. Sci.* 2021, *185*, 101–108. [CrossRef]
- Barbieri, G.; Fantuzzi, C.; Borsari, R. A model-based design methodology for the development of mechatronic systems. *Mechatronics* 2014, 24, 833–843. [CrossRef]
- 3. Alcázar-García, D.; Martínez, J.L.R. Model-based design validation and optimization of drive systems in electric, hybrid, plug-in hybrid and fuel cell vehicles. *Energy* **2022**, 254, 123719. [CrossRef]
- 4. Dini, P.; Saponara, S. Processor-in-the-Loop Validation of a Gradient Descent-Based Model Predictive Control for Assisted Driving and Obstacles Avoidance Applications. *IEEE Access* **2022**, *10*, 67958–67975. [CrossRef]
- Cosimi, F.; Dini, P.; Giannetti, S.; Petrelli, M.; Saponara, S. Analysis and design of a non-linear MPC algorithm for vehicle trajectory tracking and obstacle avoidance. In *International Conference on Applications in Electronics Pervading Industry, Environment and Society*; Springer: Cham, Switzerland, 2021; pp. 229–234.
- Bernardeschi, C.; Dini, P.; Domenici, A.; Mouhagir, A.; Palmieri, M.; Saponara, S.; Sassolas, T.; Zaourar, L. Co-simulation of a Model Predictive Control System for Automotive Applications. In *International Conference on Software Engineering and Formal Methods*; Springer: Cham, Switzerland, 2022.
- Karnopp, D.C.; Margolis, D.L.; Rosenberg, R.C. System Dynamics: Modeling, Simulation, and Control of Mechatronic Systems; John Wiley & Sons: Hoboken, NJ, USA, 2012.
- 8. Chen, X.; Salem, M.; Das, T.; Chen, X. Real time software-in-the-loop simulation for control performance validation. *Simulation* **2008**, *84*, 457–471. [CrossRef]
- NXP. NXP Model-Based Design Toolbox (MBDT). 2022. Available online: https://www.nxp.com/design/development-boards/ motor-control-development-solutions/model-based-design-toolbox-mbdt:MBDT (accessed on 10 November 2022).
- 10. Zheng, C.; Qin, X.; Eynard, B.; Li, J.; Bai, J.; Zhang, Y.; Gomes, S. Interface model-based configuration design of mechatronic systems for industrial manufacturing applications. *Robot. Comput.-Integr. Manuf.* **2019**, *59*, 373–384. [CrossRef]
- 11. Millitzer, J.; Mayer, D.; Henke, C.; Jersch, T.; Tamm, C.; Michael, J.; Ranisch, C. Recent developments in hardware-in-the-loop testing. *Model Valid. Uncertain. Quantif.* **2019**, *3*, 65–73.
- 12. SpeedGoat. Real-Time Simulation and Testing. Emulation of Plant Components. 2022. Available online: https://www.speedgoat. com/ (accessed on 10 November 2022).
- 13. Zhou, M.; Zhang, X.; Zhao, W.; Ji, J.; Hu, J. Influence of magnet shape on the cogging torque of a surface-mounted permanent magnet motor. *Chin. J. Electr. Eng.* **2019**, *5*, 40–50. [CrossRef]
- 14. Anuja, T.; Doss, M.A.N. Reduction of cogging torque in surface mounted permanent magnet brushless DC motor by adapting rotor magnetic displacement. *Energies* **2021**, *14*, 2861. [CrossRef]
- An, Y.; Ma, C.; Zhang, N.; Guo, Y.; Degano, M.; Gerada, C.; Bu, F.; Yin, X.; Li, Q.; Zhou, S. Calculation model of armature reaction magnetic field of interior permanent magnet synchronous motor with segmented skewed poles. *IEEE Trans. Energy Convers.* 2021, 37, 1115–1123. [CrossRef]
- 16. Tong, W.; Li, S.; Pan, X.; Wu, S.; Tang, R. Analytical model for cogging torque calculation in surface-mounted permanent magnet motors with rotor eccentricity and magnet defects. *IEEE Trans. Energy Convers.* **2020**, *35*, 2191–2200. [CrossRef]
- 17. Ortega, A.J.P.; Paul, S.; Islam, R.; Xu, L. Analytical model for predicting effects of manufacturing variations on cogging torque in surface-mounted permanent magnet motors. *IEEE Trans. Ind. Appl.* **2016**, *52*, 3050–3061. [CrossRef]
- 18. Hemeida, A.; Hannon, B.; Vansompel, H.; Sergeant, P. Comparison of three analytical methods for the precise calculation of cogging torque and torque ripple in axial flux PM machines. *Math. Probl. Eng.* **2016**, 2016, 2171547. [CrossRef]
- 19. Shin, K.H.; Park, H.I.; Cho, H.W.; Choi, J.Y. Analytical calculation and experimental verification of cogging torque and optimal point in permanent magnet synchronous motors. *IEEE Trans. Magn.* **2017**, *53*, 1–4. [CrossRef]
- 20. Ebadi, F.; Mardaneh, M.; Rahideh, A.; Bianchi, N. Analytical energy-based approaches for cogging torque calculation in surface-mounted PM motors. *IEEE Trans. Magn.* 2019, 55, 1–10. [CrossRef]
- Yang, Y.; Bianchi, N.; Zhang, C.; Zhu, X.; Liu, H.; Zhang, S. A method for evaluating the worst-case cogging torque under manufacturing uncertainties. *IEEE Trans. Energy Convers.* 2020, 35, 1837–1848. [CrossRef]
- 22. Kim, S.; Lee, S.G.; Kim, J.M.; Lee, T.H.; Lim, M.S. Uncertainty identification method using kriging surrogate model and Akaike information criterion for industrial electromagnetic device. *IET Sci. Meas. Technol.* **2020**, *14*, 250–258. [CrossRef]
- 23. Lee, C.S.; Kim, H.J. Harmonic Order Analysis of Cogging Torque for Interior Permanent Magnet Synchronous Motor Considering Manufacturing Disturbances. *Energies* **2022**, *15*, 2428. [CrossRef]
- 24. Verma, M.; Singh, M.; Sreejeth, M. Integrated Taguchi method-assisted polynomial Metamodelling & Genetic Algorithm based optimisation of a surface inset permanent synchronous motor for performance improvement. *IET Electr. Syst. Transp.* **2022**, *12*, 26–35.
- 25. Reales, A.; Jara, W.; Hermosilla, G.; Madariaga, C.; Tapia, J.; Bramerdorfer, G. A Machine Learning Based Method to Efficiently Analyze the Cogging Torque Under Manufacturing Tolerances. In Proceedings of the 2021 IEEE Energy Conversion Congress and Exposition (ECCE), Virtual, 10–14 October 2021; pp. 1353–1357.
- Brescia, E.; Costantino, D.; Massenio, P.R.; Monopoli, V.G.; Cupertino, F.; Cascella, G.L. A Design Method for the Cogging Torque Minimization of Permanent Magnet Machines with a Segmented Stator Core Based on ANN Surrogate Models. *Energies* 2021, 14, 1880. [CrossRef]

- 27. Vukosavic, S.N. Mathematical model of synchronous machine. In *Electrical Machines*; Springer: Berlin/Heidelberg, Germany, 2013; pp. 545–569.
- 28. Das, S. Modeling and Simulation of Mechatronic Systems Using Simscape. Synth. Lect. Mech. Eng. 2020, 5, 1–171.
- 29. Gasparetto, A.; Seriani, S.; Scalera, L. Modelling and control of mechatronic and robotic systems. *Appl. Sci.* **2021**, *11*, 3242. [CrossRef]
- 30. Wu, Y.; He, N.; Chen, M.; Xu, D. Generalized space-vector-modulation method for soft-switching three-phase inverters. *IEEE Trans. Power Electron.* **2020**, *36*, 6030–6045. [CrossRef]
- 31. Pila, A.W. Introduction To Lagrangian Dynamics; Springer: Berlin/Heidelberg, Germany, 2019.
- 32. Mora, A.; Orellana, A.; Juliet, J.; Cardenas, R. Model predictive torque control for torque ripple compensation in variable-speed PMSMs. *IEEE Trans. Ind. Electron.* **2016**, *63*, 4584–4592. [CrossRef]
- 33. Fei, Q.; Deng, Y.; Li, H.; Liu, J.; Shao, M. Speed ripple minimization of permanent magnet synchronous motor based on model predictive and iterative learning controls. *IEEE Access* **2019**, *7*, 31791–31800. [CrossRef]
- Suchỳ, O.; Janouš, Š.; Talla, J.; Peroutka, Z. Torque Ripple Reduction of IPMSM Drive with Non-sinusoidal Back-EMF by Predictive Control. In Proceedings of the 2022 IEEE 31st International Symposium on Industrial Electronics (ISIE), Anchorage, AL, USA, 1–3 June 2022; pp. 1064–1069.
- Huang, W.; Hua, W. A finite-control-set-based model-predictive-flux-control strategy with iterative learning control for torque ripple minimization of flux-switching permanent magnet machines. In Proceedings of the 2016 IEEE Vehicle Power and Propulsion Conference (VPPC), Hangzhou, China, 17–20 October 2016; pp. 1–6.
- 36. Isidori, A. Lectures in Feedback Design for Multivariable Systems; Springer: Berlin/Heidelberg, Germany, 2017.
- 37. Dini, P.; Saponara, S. Cogging torque reduction in brushless motors by a nonlinear control technique. *Energies* **2019**, *12*, 2224. [CrossRef]
- Dini, P.; Saponara, S. Design of an observer-based architecture and non-linear control algorithm for cogging torque reduction in synchronous motors. *Energies* 2020, 13, 2077. [CrossRef]
- Bernardeschi, C.; Dini, P.; Domenici, A.; Saponara, S. Co-simulation and Verification of a Non-linear Control System for Cogging Torque Reduction in Brushless Motors. In *International Conference on Software Engineering and Formal Methods*; Springer: Berlin/Heidelberg, Germany, 2019; pp. 3–19.
- 40. Dini, P.; Saponara, S. Model-Based Design of an Improved Electric Drive Controller for High-Precision Applications Based on Feedback Linearization Technique. *Electronics* **2021**, *10*, 2954. [CrossRef]
- Pierpaolo, D.; Saponara, S. Control System Design for Cogging Torque Reduction Based on Sensor-Less Architecture. In International Conference on Applications in Electronics Pervading Industry, Environment and Society; Springer: Berlin/Heidelberg, Germany, 2019; pp. 309–321.
- 42. Bernardeschi, C.; Dini, P.; Domenici, A.; Palmieri, M.; Saponara, S. Formal verification and co-simulation in the design of a synchronous motor control algorithm. *Energies* **2020**, *13*, 4057. [CrossRef]
- 43. Arias, A.; Caum, J.; Ibarra, E.; Grino, R. Reducing the cogging torque effects in hybrid stepper machines by means of resonant controllers. *IEEE Trans. Ind. Electron.* **2018**, *66*, 2603–2612. [CrossRef]
- Ignatev, E.A.; Nos, O.V. Torque Ripple Reduction of Permanent Magnet Synchronous Motor Using Proportional-Integral-Resonant Controller with Delay of Control Response Compensation. In Proceedings of the 2021 IEEE 22nd International Conference of Young Professionals in Electron Devices and Materials (EDM), Souzga, Russia, 30 June–4 July 2021; pp. 431–435.
- 45. Gao, J.; Wu, X.; Huang, S.; Zhang, W.; Xiao, L. Torque ripple minimisation of permanent magnet synchronous motor using a new proportional resonant controller. *IET Power Electron.* **2017**, *10*, 208–214. [CrossRef]
- Chuan, H.; Fazeli, S.M.; Wu, Z.; Burke, R. Mitigating the torque ripple in electric traction using proportional integral resonant controller. *IEEE Trans. Veh. Technol.* 2020, 69, 10820–10831. [CrossRef]
- Rafaq, M.S.; Midgley, W.; Steffen, T. Proportional-integral-resonant control for the periodic disturbance minimization of the PMSM. In Proceedings of the 11th International Conference on Power Electronics, Machines and Drives (PEMD 2022), Newcastle, UK, 21–23 June 2022; pp. 240–244.
- Minghe, T.; Bo, W.; Yong, Y.; Xing, M.; Qinghua, D.; Dianguo, X. Proportional resonant-based active disturbance rejection control for speed fluctuation suppression of PMSM drives. In Proceedings of the 2019 22nd International Conference on Electrical Machines and Systems (ICEMS), Harbin, China, 11–14 August 2019; pp. 1–6.
- 49. Dini, P.; Saponara, S. Electro-Thermal Model-Based Design of Bidirectional On-Board Chargers in Hybrid and Full Electric Vehicles. *Electronics* **2021**, *11*, 112. [CrossRef]
- Benedetti, D.; Agnelli, J.; Gagliardi, A.; Dini, P.; Saponara, S. Design of an Off-Grid Photovoltaic Carport for a Full Electric Vehicle Recharging. In Proceedings of the 2020 IEEE International Conference on Environment and Electrical Engineering and 2020 IEEE Industrial and Commercial Power Systems Europe (EEEIC/I&CPS Europe), Madrid, Spain, 9–12 June 2020; pp. 1–6.
- Benedetti, D.; Agnelli, J.; Gagliardi, A.; Dini, P.; Saponara, S. Design of a digital dashboard on low-cost embedded platform in a fully electric vehicle. In Proceedings of the 2020 IEEE International Conference on Environment and Electrical Engineering and 2020 IEEE Industrial and Commercial Power Systems Europe (EEEIC/I&CPS Europe), Madrid, Spain, 9–12 June 2020; pp. 1–5.
- 52. Hu, Y.; Gu, W.; Zhang, H.; Chen, H. Adaptive robust triple-step control for compensating cogging torque and model uncertainty in a DC motor. *IEEE Trans. Syst. Man, Cybern. Syst.* 2018, 49, 2396–2405. [CrossRef]

- Huang, W.; Hua, W.; Zhu, X.; Fan, Y.; Cheng, M. Comparison of Cogging Torque Compensation Methods for a Flux-Switching Permanent Magnet Motor by Harmonic Current Injection and Iterative Learning Control. In Proceedings of the 2020 International Conference on Electrical Machines (ICEM), Gothenburg, Sweden, 23–26 August 2020; Volume 1, pp. 1971–1977.
- Dou, B.; He, R.; Wang, Z.; Li, Q.; Chen, C.; Dong, J.; Peng, Q.; Yu, T.; Huang, Y. The Repetitive Controller Design for Inhibiting the Torque Ripple of PMSM. In Proceedings of the 2020 7th International Conference on Information Science and Control Engineering (ICISCE), Changsha, China, 18–20 December 2020; pp. 2245–2249.
- 55. Dini, P.; Saponara, S. Design of adaptive controller exploiting learning concepts applied to a BLDC-based drive system. *Energies* **2020**, *13*, 2512. [CrossRef]
- 56. Liu, J.; Li, H.; Deng, Y. Torque ripple minimization of PMSM based on robust ILC via adaptive sliding mode control. *IEEE Trans. Power Electron.* **2017**, *33*, 3655–3671. [CrossRef]
- Du, R.; Li, C.; Bu, F.; Liu, J. Low-Speed Disturbance Suppression Strategy of Direct Drive Servo Motor Based on Combination of Sliding Mode Control and Disturbance Compensation. In Proceedings of the 2021 24th International Conference on Electrical Machines and Systems (ICEMS), Gyeongju, Korea, 31 October–3 November 2021; pp. 1933–1937.